Editor comments:

1. Please clarify whether informed consent was required for this study. If so, was it obtained?

Informed consent and REB was used for the original study of the OSCE.

See page 9, line 7: All participants consented to the use of their de-identified data.
2. Please add all aus to Authors' Contribution (missing DW)

Thank you, this has been done.

3. Please include the abstract following the title page

Thank you, this has been done. See page 3, line 6.